ARTICLE

https://doi.org/10.1057/s41599-023-02527-8

OPEN

Check for updates

Outperforming the market: a comparison of Star and NonStar analysts' investment strategies and recommendations

Darko B. Vukovic^{1,2™}, Orifjon O. U. Kurbonov³, Moinak Maiti₁⁰⁴, Mustafa Özer⁵ & Milan Radovanovic²

We employ StarMine to investigate the impact of analyst recommendations on stock performance. We test whether star-ranked analysts generate abnormal returns and outperform non-stars in short and long portfolios. Utilizing buy-and-hold calendar-time portfolio methodology, we calculate portfolio alphas using various asset pricing models, including CPM, the Fama and French 3-factor model, and the Carhart 4-factor model. Results indicate that all analyst groups can generate abnormal returns exceeding the market average. Star-ranked analysts outperform non-stars in short portfolios by 0.5523% in monthly alpha, though no significant difference exists in long portfolio alphas. We also conduct regressor endogeneity tests and explore investor sentiment mechanisms by utilizing the GARCH model and frequency-domain causality analysis, with NASDAQ as a proxy for investor sentiment. These tests reveal that the momentum factor is exogenous, and investor sentiments have a statistically significant positive effect on stock return volatility, with changes occurring between 5 and 10 days. This research underscores the value of analyst insights for investors, validates StarMine's ranking effectiveness, and suggests market participants can benefit from incorporating analyst recommendations into their investment decisions. Our study makes a significant contribution to the existing literature by introducing a novel approach to understanding investor sentiment mechanisms through a causality model.

¹Graduate School of Management, Saint Petersburg State University, Volkhovskiy Pereulok 3, 199004 Saint Petersburg, Russia. ² Geographical Institute "Jovan Cvijic" SASA, Djure Jaksica 9, 11000 Belgrade, Serbia. ³ School of Accounting and Finance, University of Vaasa, Wolffintie 34, 65200 Vaasa, Finland. ⁴ Independent Researcher, Kolkata 711112, India. ⁵ Department of Economics, Faculty of Economics and Administrative Sciences, Anadolu University, Eskisehir, Turkey. [⊠]email: d.vukovic@gi.sanu.ac.rs; vdarko@hotmail.rs

Introduction

he efficient market hypothesis (EMH) by Fama (1965) states that if stock prices incorporate all available information in the market, it becomes challenging to leverage information for profit in such a highly efficient market (Li et al. 2017a). This process is explained by random walk, where changes in stock price are random. When new information arrives in the market, market participants are promptly informed and act upon the arrival of this new information. Several studies (De Bondt and Thaler, 1985; Brown et al. 1993) suggest that this new information could introduce uncertainty, leading to potential overreactions by investors who may sell or buy stocks in order to rebalance their investment portfolios. According to De Bondt and Thaler (1985) and Brown et al. (1993), this new information could lead to overreactions, prompting investors to sell or buy stocks to rebalance their investment portfolios. Thus, investors do react to the arrival of new information, which directly impacts the stock's performance. However, this explanation pertains to market reactions and does not directly challenge the EMH. The signaling theory suggests that analyst recommendations are useful in judging the underlying firm's value (Yasar et al. 2020). Any good news associated with the concerned firm may drive its stock prices. Similarly, bad news may drive its stock prices downward. It all suggests that analyst recommendations can influence stock performance due to information asymmetry and affect market efficiency.

The studies of Stickel (1995) and Womack (1996) argue that analysts' recommendations can potentially generate positive/ negative market responses. Other studies also affirm that upside/ downside changes in analysts' recommendations can be a source of positive or negative price drifts at the time of announcement (Barber et al. 2010; Vukovic et al. 2021 and 2020). With some evidence that analysts' recommendations do have some investment value, numerous studies have attempted to determine how the stock market responds to analysts with different attributes. Few studies examine the reputation of analysts and the success or failure of their stock-picking. (Emery and Li, 2009; Fang and Yasuda, 2009, 2014; Kucheev and Sorensson, 2019). The study of Li et al. (2017b) indicate that analysts can play a vital role in shaping the rival responses of target firms. Beside this, in some recent studies, authors utilize turnover rates to characterize the sentiment of investors (Liu et al. 2019).

The aim of this study is to explore whether the star status of an analyst can generate abnormal returns. Although there has been a large amount of research on the analysts' reputation (Fang and Yasuda, 2014; Kucheev and Sorensson, 2019; Su et al. 2019) and market reactions, many of the findings have been controversial. Moreover, to the best of the study's knowledge, the majority of scholars use Institutional Investor Magazine rankings data to determine the star status of an analyst. Analysts with "star" status have achieved a high level of success in providing valuable insights and generating excess returns through their stock recommendations.

The present study has the following characteristics: First, the study examines the NASDAQ market, where participants trade through a dealer as opposed to directly with one another (as in the case of an auction market, like the NYSE). It is a modified capitalization-weighted index, and this approach allows for more diversification. Second, the present study use StarMine analyst rankings, while most of the earlier studies analyze Institutional Investor Magazine analyst rankings. The Institutional Investor Magazine analyst rankings are determined by institutional investors' votes and input. StarMine analyst rankings, on the other hand, are based on a proprietary quantitative algorithm. As a result, StarMine analyst rankings are calculated using a more quantitative and data-driven approach. Third, unlike Kucheev

and Sorensson (2019), the analysis does not divide the portfolios into two time periods depending on the year in which the analyst becomes a star and the year after their star status. Instead, it implies that experts who have been named stars in the past are more likely to make greater recommendations and are evaluated appropriately. Fourth, the study performs a further robustness test for analysis as follows: Star analysts are ranked first in Star-Mine's rankings are excluded from the initial sample of Star analysts. In light of this, we postulate two research questions:

RQ1: Can investors use analyst recommendations as an investment strategy to earn above-average returns? RQ2: Therefore, would analysts with a better reputation make more profitable recommendations than those with a lower reputation? The novelty of our study is in addition to the regressor endogeneity testing and the investor sentiment mechanism's employment with causality analysis. After portfolio construction, we calculate the risk-adjusted returns using multifactor asset pricing models with CAPM, Fama-French 3-factor, and Carhart 4-factor models. For testing the regressor endogeneity issue, we use the Durbin-Wu-Hausman specification test. Further, our study investigates the precise model through which star analysts' influence impacts fluctuations in stock prices. For this purpose, we employ the investor sentiment mechanism, where the NASDAQ Volatility Index (VXN) proxy is utilized as an exogenous regressor in the variance equation. Lastly, to examine the prevalence of the impact of investor sentiment on stock market volatility, we utilize the frequency-domain causality test.

Our study makes a significant contribution to the existing literature by introducing a novel approach to understanding investor sentiment mechanisms through a causality model. To the best of our knowledge, this is the first study that analyzes a proxy for investor sentiment and tests its statistically significant impact on stock price volatility. To achieve this, we employ the GARCH model and frequency-domain causality analysis, utilizing NAS-DAQ as a proxy for investor sentiment. In addition, our study addresses the issue of endogeneity, as examined by the robust Durbin-Wu-Hausman model. Furthermore, we contribute by providing unprecedented evidence regarding the influence of investor sentiment while distinguishing between temporary and permanent effects. Our findings reveal that investor sentiment is statistically significant and exerts a positive impact on stock return volatility. Notably, our study uncovers that stock market volatility induces changes in investor sentiment within a 5 to 10day timeframe, with effects that are not of a permanent nature. This observation marks a valuable addition to the existing body of literature.

The remainder of the paper continues as follows: Section 2 analyzes and synthesizes available literature on analysts' recommendations and reputations; Section 3 covers the description of the dataset and the methodology; Section 4 discusses the key findings from the empirical results; and Section 5 concludes and offers suggestions for further research.

Literature background

Investment value of recommendations. Since Stickel (1995) and Womack (1996), the literature on the economic value of analyst opinions has significantly expanded (1996). According to Womack (1996), upgrades trigger a 30-day stock price drift, while price trends resulting from downgrades can persist for up to six months. Stickel (1995) investigates the relationship between analyst reputation and short-term price reactions based on nearly 17,000 recommendations from various analysts (proxied by their ranking in Institutional Investor Magazine's All America Research Team). The study reveals a positive relationship

between analyst recommendations and short-term price reactions over a 12-year period.

However, subsequent studies have found that price reactions to analysts' recommendations are often short-lived and do not translate into profitable investment strategies (Jegadeesh and Kim, 2006; Leone and Wu, 2007; Barber et al. 2010; Loh and Stulz, 2011; Su et al. 2019). Leone and Wu (2007) report a positive relationship between analysts' recommendations and stock performance, especially for analysts with an AA status, indicating a persistent relationship. In contrast, Su et al. (2019) find that, assuming no transaction costs, upgrade recommendations fail to generate significant excess returns, while downgrade recommendations can result in large abnormal returns. Hobbs et al. (2021) analyze data from 1994 to 2015 to identify institutional investors managing more than \$100 million who rarely rely on sell-side reports as an example of market participant behavior. Their study demonstrates that when institutional investors trade against recent analyst recommendation changes, they tend to incur losses.

Analysts' reputation and stock performance. Following Stickel's seminal work in 1995, there has been a substantial increase in scholarly investigations focusing on the relationship between analyst reputation and stock performance (Leone and Wu, 2007; Emery and Li, 2009; Loh and Stulz, 2011; Fang and Yasuda, 2014; Kucheev and Sorensson, 2019; Byun and Roland, 2020). Most studies initially examined the average price reactions, but recent research has shifted its attention to assessing the profitability by comparing the performance of star-rated and non-star-rated entities. Emery and Li (2009), for instance, compared the stock performance and determinants of rankings using a dataset of 20,239 recommendations issued by nearly 6000 analysts between 1993 and 2005. Their study contrasted Institutional Investor Magazine (II) and Wall Street Journal (WSJ) rankings, revealing that factors related to recognition play a significant role in determining rankings. Several studies argue that II rankings may be biased (Kessler, 2001; Kucheev and Sorensson, 2019).

Guo et al. (2020) contend that analyst recommendations often carry biases that create market friction. However, both Su et al. (2019) and Kucheev and Sorensson (2019) establish a wellestablished positive correlation between analyst reputation and their recommendation performance. Yet, some argue that an analyst's reputation might not solely stem from their skill but could be influenced by the brokerage house's reputation within which they work. Analysts often utilize in-house resources, influencing their recommendations. Fang and Yasuda (2014) even suggest that some top-ranked analysts in II may have achieved their status through fortunate circumstances. Kadan et al. (2020) lend credence to these perspectives by suggesting that analysts' stock selection proficiency and prediction accuracy depend on specific benchmarks. Fang and Yasuda (2014) categorized stock recommendations into reputation groups using the analyst's placement in the II All-America Research Team as a proxy, utilizing a substantial dataset spanning from 1993 to 2009.

To evaluate the comparative performance of star and non-star analysts, our study calculates risk-adjusted returns using various models, including the Capital Asset Pricing Model (CAPM) proposed by Sharpe (1964), the Fama-French 3-factor model introduced by Fama and French (1993), the Carhart 4-factor model developed by Carhart (1997), and a 5-factor model that includes a tech-sector index return. Barber et al. (2007) find that the performance of star analysts exhibits persistence, indicating that it's not solely due to chance but rather attributed to their enhanced access to firm management. Kucheev and Sorensson (2019) conduct a comparative analysis of different ranking systems, yielding the most significant average excess returns using a long-short portfolio strategy. Su et al. (2019) focus on the UK stock market and the reputation effect inside brokerage houses. Byun and Roland (2020) raise concerns about the suitability of forecast attributes and rankings data in previous investigations, suggesting that market participants assign greater significance to large organizations. This trend may contribute to the inconsistent findings observed in research projects employing II rankings. Kadan et al. (2020) confirm this by demonstrating that Fama-French factor alphas have statistical significance for small and low-coverage stocks but lack significance for large and high-coverage stocks.

Analysts' ranking systems. The measurement of analysts' reputation and its impact on stock performance has been a subject of extensive scholarly discourse. Within this literature, there has been considerable deliberation over the appropriate methods for measuring reputation and identifying the measurement types that yield the least biased outcomes. Table 1 illustrates the primary classifications of rankings in literature. Based on the evaluation methodology, it is possible to categorize each rating into two distinct types: objective rankings, such as StarMine and WSJ, and subjective rankings, exemplified by Institutional Investor Magazine. In summary, subjective rankings employ various evaluation methodologies and are grounded in survey outcomes. Two reputable and unbiased objective rankings are the "Best on the Street" published by the Wall Street Journal and "Top Stock Pickers" published by StarMine. The third objective ranking pertains to StarMine's "Top Earnings Estimators," which evaluates the precision and timeliness of analysts' earnings projections.

Rankings provided by Institutional Investor Magazine (II) are considered subjective, as they are susceptible to influences similar to "popularity contests" (Emery and Li, 2009). The evaluation process involves II distributing a survey to buy-side managers, soliciting their assessments of performance and several aspects related to sell-side analysts.

Kucheev and Sorensson (2019) argue that existing rankings inadequately assess portfolio profitability. They also provide evidence of significant alphas for the cluster of star analysts from II rankings, though with less informative relevance compared to rankings from StarMine and the Wall Street Journal. Byun and Roland (2020) conduct a critical examination of the subjectivity in rankings and raise concerns about market players' limited focus on major enterprises. The Wall Street Journal (WSJ) issues rankings known as "Best on the Street." This ranking system is typically determined by a cumulative score, derived from analysts' assessments over the previous year and computed as the total of one-day returns associated with their recommendations (Emery and Li, 2009). The ranking system has a short-term focus as it prioritizes analysts who offer recommendations on the same day as a substantial price fluctuation, while analysts who issue recommendations before or after the significant price change are penalized. These WSJ rankings may introduce substantial random influences in the selection process of prominent individuals. Emery and Li (2009) also document the significant underperformance of WSJ stars following their election, compared to those who were not designated as stars."

Thomson Reuters' StarMine rankings, which include "Top Stock Pickers" (TSP) and "Top Earnings Estimators" (TEE), have been issued annually since 1998. They are typically released in October each year, with exceptions in December 2009, May 2012, and August 2013. Despite their later introduction, StarMine rankings have gained significant influence in the sell-side research industry, serving as a crucial reference. Many Wall Street firms use these rankings to determine analyst compensation, as noted by Ertimur et al. (2011). Despite the relatively late introduction of

Table 1 Descriptior	ı of rankings.					
Ranking name	Rating agency	Abbreviation used in this paper	Type of ranking	Measure	Measurement	Number of analysts per industry
"All-America Research Team"	Institutional Investor	=	Subjective / Qualitative	12 criteria (most important: industry knowledge and integrity; least important: stock picking, and accuracy of EPS)	Survey	3 + Runners-up
"Top Earnings Estimators"	StarMine	TEE	Objective ∕ Quantitative	Accuracy and timing of earnings estimations	Calculation - EPS	m
"Top Stock Pickers"	StarMine	TSP	Objective ∕ Ouantitative	Excess returns on individual portfolios	Calculation - Recommendations	m
"Best on the Street"	The Wall Street Journal	WSJ	Objective / Quantitative	Total score for stock returns	Calculation - Recommendations	5 in 2003-2011, 3 in 2012, 2013
Table 1 shows the key differer nature. "Top Earnings Estimet II chooses 3 analysts per indu: excess returns obtained by fol did not publish the analysts '	cces between the measures. Rankings a cors" and "Top Stock Pickers" by StarMi stry and "runners-up"—those who will I llowing the analysts' recommendations rankings. For more details, the study b	re divided into two groups (type of ranking ine (Refinitiv) and "Best on the Street" by 1 potentially be chosen as stars in future yee s. Both the TSP and TEE choose three analy by Kucheev et al. (2017) may be referred	c) according to the ranking's evaluatic fine Wall Street Journal are in the objic res. TEE and TSP have a substantial di usts per industry every year. Similar to to.	in approach. "All-America Research Team," issued by ective group. II rankings assess the analysts based on fference in measurement type. While TEE represents 5 TSP, WSJ rankings are also based on the recomme	Institutional Investor Magazine, is in the subjec 12 various criteria, where stock-picking skills are the accuracy and timing of EPS calculations, T51 adations and the total return of the recommenden adations and the total return of the recommenden.	tive group due to its survey not the foremost criterion. ⁹ is calculated based on the d stock. For 2021, the WSJ

StarMine rankings, they hold a significant role in analysts' research, providing a vital and influential reference in the industry (Kim and Zapatero, 2023).

The determination of the TSP ranking is based on the abnormal returns generated by an unleveraged portfolio constructed according to analysts' recommendations. Calculating analysts' returns involves creating both long and short portfolios using a buy-and-hold approach, considering the market capitalization-weighted portfolio of all companies within a specific industry. Rebalancing occurs on a monthly basis and in response to an analyst's recommendation revision or the inclusion/exclusion of coverage.

The TEE ranking assesses how accurately each analyst predicts future earnings. As each analyst is evaluated relative to their peers, the TEE ranking can be seen as a relative accuracy measurement tool. This measure encompasses various factors, including the analyst's forecast error, the variance of the error, the analyst's error compared to other analysts' errors, the absolute earnings value of the firm, and the timing of measurements. The rankings are calculated on a daily basis, taking into account scores assigned to stocks, industries, and analysts. Unlike TSP and WSJ rankings, which give more weight to the investment value of analysts' recommendations, TEE rankings place greater emphasis on earnings forecasts (Kucheev and Sorensson, 2019).

Data and methods

Data and sample. We utilize multiple sources for data extraction. Stock recommendations are sourced from the Thomson Financials Institutional Brokers' Estimate System (I/B/E/S) Detailed Recommendations file. This system standardizes the diverse brokers' recommendation systems into a 5-point scale, where 1 corresponds to "strong buy," 2 to "buy," 3 to "hold," 4 to "sell," and 5 to "strong sell." Given the potential for different brokers to use varying measurement scales, standardization is of paramount importance. I/B/ E/S Estimates, a component of the Institutional Brokers' Estimate System, has served as the industry standard for collecting and consolidating analyst forecasts of future earnings for publicly traded companies since its establishment in 1976. The dataset encompasses a vast geographical reach, including data from over 23,400 active companies across more than 90 countries, sourced from a network of 950+ firms and over 19,000 analysts, ranging from global to regional and local brokers. Notably, this data source stands out due to its extensive historical database, dating back to 1976 for North American data and 1987 for international data. It offers a range of data formats (Refinitiv, 2023). The daily holding period returns (HPR) for a stock, including dividends and other price adjustments like splits, are retrieved from the comparable Thomson Reuters' Refinitiv Eikon. In the realm of financial analysis, HPR refers to the overall return on an investment during the time it is held. It offers a comprehensive perspective on the performance of a stock, considering both capital appreciations, reflected in changes in the stock price, and income generated from dividends. Additionally, HPR accounts for various corporate actions, such as stock splits, which can impact the stock's price dynamics. The use of Thomson Reuters' Refinitiv Eikon dataset ensures the availability of accurate and reliable daily HPR data, providing a robust foundation for our empirical analysis. The value-weighted market return, book-tomarket, size, and momentum factors are obtained from the Fama-French factors daily frequency database and used for regression analysis.¹. The database is a reputable source that has gained widespread acceptance and is extensively used in finance research. Researchers depend on it for conducting robust risk-adjusted analyses, thanks to its consistency and transparency in financial research (see, for example, Hou et al. 2015; Jegadeesh and Titman, 1993; Carhart, 1997). Importantly, for each year of the analysis

Table 2 Recommendation sample described by year and number of analysts on cov

Year	Firms	Analysts	Stars (%)	Non-Stars (%)	Recommendation	Stars (%)	Non-Stars (%)
2010	88	424	3.1%	96.9%	816	2.6%	97.4%
2011	86	385	3.4%	96.6%	725	3.3%	96.7%
2012	93	373	1.6%	98.4%	656	1.7%	98.3%
2013	92	326	2.1%	97.9%	606	1.8%	98.2%
2014	92	307	2.9%	97.1%	502	2.6%	97.4%
2015	93	310	2.3%	97.7%	562	1.8%	98.2%
2016	96	296	3.4%	96.6%	601	1.7%	98.3%
2017	97	338	2.4%	97.6%	667	1.6%	98.4%
2018	94	213	1.4%	98.6%	406	1.0%	99.0%
2019	90	231	0.4%	99.6%	485	0.2%	99.8%
2020	98	246	1.6%	98.4%	561	0.7%	99.3%
Average	93	314	2%	98%	599	2%	98%

The table presents the number of firms, analysts, and recommendations included in the sample from 2010 to 2020 on a yearly basis. The left part of the table shows the number of analysts and the share (%) of Star and Non-Star analysts overall. The right part of the table indicates the overall number of recommendations and the share (%) of recommendations made by Star and Non-Star analysts. The number of recommendations does not include the reiterations but only the level changes.

(2010-2020), data on the analysts' ranks has been extracted by hand from StarMine.

StarMine's ranking data is integrated with the recommendation file using the analyst's name, the broker's affiliation, and the industry codes as reconciliation points. Ensuring each analyst is accurately linked to their recommendations requires meticulous work and double verification. As previously mentioned, StarMine annually releases a list of the top 1–3 analysts within each GICS industry, designating the top analyst as a "super-star" in their respective field (StarMine, 2020).

During data cleaning, it is important to note that the I/B/E/S recommendation file includes recommendations from anonymous analysts or those lacking relevant industry or brokerage house codes. Since these details are irrelevant to the study, they are excluded from the dataset before analysis (Kucheev and Sorensson, 2019). Additionally, stock recommendations tend to remain static and mature over time, diminishing their usefulness (Barber et al. 2010; Jagadeesh et al., 2014). Therefore, this research considers only shifts in recommendations, such as from "Buy" to "Sell" or "Sell" to "Buy," and initial recommendations (when the analyst starts covering), while disregarding subsequent occurrences of the same recommendation level. We combine data from two analyst rankings, TEE and TSP rankings, to create a single variable indicating an analyst's star status. This process begins by identifying the analysts who hold star status and those who do not. The primary objective of this combination is to expand the pool of star analysts. The underlying assumption is that both types of rankings offer insights into exceptionally skilled analysts. Therefore, our study categorizes a group of analysts into two distinct categories: "star" and "non-star." Table 2 presents the summary statistics for the recommendation sample. Each year, the dataset encompasses approximately 90 to 100 companies, which are constituents of the NASDAQ 100 index. We specifically concentrate on NASDAQ 100 companies for a specific reason: we intend to investigate the impact of excess returns within the highly dynamic tech sector. While many studies utilize the broader S&P index, we acknowledge that tech companies, characterized by their rapid responses to market shifts, provide a distinct environment for this analysis. This approach enables us to gain insights into the unique dynamics of the tech sector and their significance in the realm of financial research and investment strategies. The number of recommendations show the same trend as analysts, declining from 816 in 2010 to 561 in 2020. The number of recommendations issued by stars is at 2% on average, while non-stars issue almost 98% of all recommendations.

Table 3 presents the 5-level recommendations issued by different groups of analysts over the years from 2010 to 2020. Interestingly, in both analyst groups, the number of "hold" recommendations holds the majority, comprising approximately 41% of the total for both groups. On the other hand, "sell" and "strong sell" recommendations constitute the smallest portion of the sample. Among non-stars, the average percentage of "sell" recommendations is approximately 3.8%, while "strong sell" recommendations account for 1.3%. The Star analyst group has no observations for the "strong sell" level, but they have 3.7% for the "sell" level. However, it's worth noting a sharp declining trend in the "sell" level among Star analysts, which dropped to 0% after 2013.

The final dataset comprises a total of 6587 observations, encompassing changes in analyst recommendation levels and coverage initiations (initial recommendations) for 93 NASDAQ 100 index companies between 2010 and 2020. Table 4 categorizes the sample of recommendations into various transition categories, with the initial negative and first positive recommendations being the only pair present in the sample, indicating that no additional recommendation levels existed before 2010. These two sets of recommendations, respectively. The trend in dynamics is negative, as evidenced by the declining number of total recommendations.

Portfolio Construction. Our study uses the portfolio construction method as described in Barber et al. (2006), and Fang and Yasuda (2014) based on analysts' recommendations (Strong Buy, Buy, Hold, Sell, Strong Sell)². We construct calendar-time portfolios for each group of analysts in the year when the analyst was elected as a Star. The approach used in our portfolio construction differs slightly from that of Kucheev and Sorensson (2019) in that we do not partition the portfolios into two distinct time periods: one following the year in which the analyst achieved star status, and the year immediately after. Instead, it assumes that everelected Star analysts are prone to make better recommendations and are tested accordingly.

Crucially, there are no strict rules of portfolio formation, meaning that some studies construct three "long,", "hold," and "short" portfolios (Kucheev and Sorensson, 2019), while others build only "long" and "short" portfolios, including "hold" recommendations in the latter category (Barber et al. 2007; Fang and Yasuda, 2014; Su et al. 2019). Long and short portfolios within each analyst cohort indicate investing strategies in which

Table 3 Percentage of each recommendation level grouped by analyst Star status.

	Non-Star	Non-Star					Star			
Year	Strong Buy	Buy	Hold	Sell	Strong Sell	Strong Buy	Buy	Hold	Sell	Strong Sell
2010	22.5%	28.2%	41.3%	4.7%	3.4%	28.6%	9.5%	52.4%	9.5%	0%
2011	21.1%	33.7%	39.2%	4.7%	1.3%	12.5%	33.3%	41.7%	12.5%	0%
2012	12.4%	38.3%	43.7%	4.8%	0.8%	9.1%	27.3%	45.5%	18.2%	0%
2013	14.5%	35.8%	45.4%	3%	1.3%	9.1%	27.3%	63.6%	0%	0%
2014	19.2%	39.3%	38.2%	2.7%	0.6%	15.4%	23.1%	61.5%	0%	0%
2015	20.1%	34.2%	42.6%	2.2%	0.9%	20%	30%	50%	0%	0%
2016	14.2%	35.4%	44%	4.7%	1.7%	10%	20%	70%	0%	0%
2017	16.6%	43.4%	37.0%	2.4%	0.5%	27.3%	27.3%	45.5%	0%	0%
2018	23.9%	27.4%	44.3%	3%	1.5%	50%	25.0%	25%	0%	0%
2019	17.6%	40.1%	38.2%	2.9%	1.2%	100%	0%	0%	0%	0%
2020	13.6%	42.4%	36.4%	6.3%	1.3%	25%	75%	0%	0%	0%
Average	17.8%	36.2%	40.9%	3.8%	1.3%	27.9%	27.1%	41.4%	3.7%	0.0%

The table illustrates the number of recommendations (strong buy, buy, hold, sell, and strong sell) from 2010 to 2020 on a yearly basis. The percentage of each recommendation level is grouped according to the star status of an analyst, where the left side of the table represents the non-star analysts, and the right side of the table represents the star analysts. Obviously, star analysts make more positive recommendations and do not make any significantly negative recommendations (strong sells) compared to non-star analysts. A substantial part of the recommendations in both analyst groups account for hold recommendations (40% on average).

Table 4 The number	of recommendation	level changes
(revisions).		

Year	First negative	First positive	From negative to positive	From positive to negative
2010	340	350	61	65
2011	208	224	171	122
2012	190	215	116	135
2013	171	174	129	132
2014	127	179	112	84
2015	137	198	107	120
2016	188	179	117	117
2017	145	266	134	122
2018	116	124	85	81
2019	133	199	81	72
2020	179	217	96	69
Total	1934	2325	1209	1119
From neaat	ive to positive and fr	om positive to negat	ive are the recommendation	on revisions from the

From negative to positive and from positive to negative are the recommendation revisions from the previous levels. In total, there are 6587 recommendations, revisions and coverage initiations observed in the sample. Coverage initiations have the largest part of the sample with 1934 negative and 2325 positive recommendations.

analysts or investors take long (buy) positions in selected assets while concurrently adopting short (sell) holdings in other assets. These tactics are utilized with the purpose of potentially capitalizing on the relative performance of assets within each portfolio. For instance, within the context of each analyst cohort, a long portfolio may contain stocks that analysts approve as "buy" recommendations, predicting that these equities will appreciate in value. Conversely, a short portfolio may consist of companies marked as "sell" recommendations by analysts, with the idea that these securities will decline in value. The cumulative performance of these long and short positions influences the overall profitability of the portfolio. The portfolio formation matrix can be seen in Table 5, where the number of recommendations in each portfolio is normally balanced and excludes reiterations.

The methodology assumes that for each new recommendation, a hypothetical investor invests \$1 at the end of the recommendation disclosure day into the matching portfolio. If the recommendation arrives on a weekend or holiday, the \$1 is invested on the following working day. If the stock is recommended by more than one analyst, it will appear several times in the corresponding portfolio. If the recommendation is "Hold", "Sell", or "Strong

Table 5 The portfolio formation matrix.

Recommendations	Long Portfolio	Short Portfolio	Total
1 - Strong Buy	1169	-	1169
2 - Buy	2365	-	2365
3 - Hold	-	2708	2708
4 - Sell	-	256	256
5 - Strong Sell	-	89	89
Total	3534	3053	6587
The "Long" Portfolio comprises	s only positive recommenda	tions ("Strong Buy"	and "Buy"), while

the "Short" Portfolio includes negative recommendations plus the neutral "Hold" recommendation.

Sell", the investor is assumed to shorten the stock by \$1 and record the investment in the "Short" portfolio. In our study, each \$1 invested is held for 30 days, assuming that a hypothetical investor is oriented toward short-term results. After 30 days from the recommendation date, each position is closed with a positive or negative return. As an additional test, Star analysts who are ranked as number one in *StarMine*'s rankings are detached from the initial sample of Star analysts.

Models. The next phase involves computing portfolio returns after the portfolios have been constructed. We employ equal monetary investment methodology which assumes that for each recommendation n, let $x_{n,t-1}$ be the compounded daily return of stock $i_{n,t}$ from the next day, when the recommendation is issued up to a future date t-1 (one day prior to date t), which can be described in the following equation (Barber et al. 2006; Fang and Yasuda, 2014; Kucheev and Sorensson, 2019):

$$x_{n,t-1} = R_{i_n, recdat_n+1} R_{i_n, recdat_n+2} * \dots * R_{i_n, recdat_nt-1}$$
(1)

where $R_{i_n,recdat_nt-1}$ is the total return of stock $i_{n,t}$ on calendar date t-1. Applying the Eq. (1) for all recommendations in the portfolio brings to the next step of daily portfolio return calculation. As such, the calendar date t gross return on a certain portfolio ρ , which contains recommendations from n = 1 to $N_{\rho t}$ will be

defined as:

$$R_{\rho t} = \left(\sum_{i=1}^{N_{\rho t}} x_{n,t-1} * R_{i_n,t}\right) / \sum_{i=1}^{N_{\rho t}} x_{n,t-1}$$
(2)

where, $N_{\rho t}$ is defined as the total number of recommendations, which appear in the corresponding portfolio ρ on date *t*. Since the equal-weighted portfolio return calculation has met strong criticism for the possible bias (Dutta, 2015; Kothari et al. 2016), our study considers the value-weighted return of the portfolio (2) ρ on date *t*-1, denoting $x_{n,t-1}$ as a weight of each recommendation *n* in long or short portfolio. When applied, the equation above yields daily time series returns for each portfolio from 2010 to 2020.

The next step assumes calculation of the risk-adjusted returns using multifactor asset pricing models such as CAPM, Fama-French 3-factor model, and the Carhart 4-factor model. In our multifactor asset pricing models, we incorporate the following factors:

- 1. The portfolio returns.
- 2. The market return of firms traded on NASDAQ.
- 3. The risk-free rate of return is based on the 1-month treasury bill.
- 4. A size factor, calculated as the difference between the returns of value-weighted portfolios comprising small and large stocks (Sun et al. 2020).
- 5. A B/M (Book-to-Market) factor, computed as the difference between the returns of value-weighted portfolios containing high and low B/M stocks.
- 6. A momentum factor, determined as the average return of two high-return portfolios minus the average return of two low-return portfolios (Vukovic et al. 2023).

The following three equations represent these 3 asset pricing models, respectively:

$$R_{\rho t} - R_{f t} = \alpha_p + \beta_p \left(R_{m,t} - R_{f,t} \right) + \varepsilon_{p,t}$$
(3)

$$R_{\rho t} - R_{f t} = \alpha_p + \beta_p \left(R_{m,t} - R_{f,t} \right) + s_p SMB_t + h_p HML_t + \varepsilon_{p,t}$$
(4)

$$R_{\rho t} - R_{ft} = \alpha_p + \beta_p \left(R_{m,t} - R_{f,t} \right) + s_p SMB_t + h_p HML_t + m_p MOM_t + \varepsilon_{p,t}$$
(5)

Where $R_{\rho t}$ is the portfolio ρ return on a certain date *t*; $R_{m,t}$ is the market return of firms traded in NASDAQ; $R_{f,t}$ is the risk-free rate of return for the 1-month treasury bill; SMB_t is a size factor, HML_t is a B/M factor, MOM_t is a momentum factor. Initially, a basic CAPM model is used to evaluate the excess return against the market factor. Supplementary, three additional factors from the Fama-French 3 factor model are integrated to examine the persistence of excess return amid additional anomalies. Finally, the fourth momentum factor from the Carhart 4-factor model is introduced to determine if the excess return remains unaffected. This methodology aligns with common practices in recommendations research and aims to ensure the stability and reliability of the results. For detail on construction of the Fama French factors (SMB, HML, and MOM) following studies may be referred (Fama and French, 1993 & 2012; Balakrishnan and Maiti, 2017; Maiti and Balakrishnan, 2018 and 2020; Maiti, 2019). Average daily excess returns are multiplied by 21 working days to obtain an excess return monthly.

Endogeneity Issue. The assumptions of contemporaneous exogeneity, often assumed but seldom verified in financial panel data models, hold significant importance. When these assumptions are violated, it can result in inconsistent parameter estimates, unreliable standard errors, and hypothesis tests that lack validity. The Durbin-Wu-Hausman specification test used assess the endogeneity issue of the variable under investigation has a wellestablished in applied studies (Mulligan, 1996; Chen and Xia, 2020; Margues, 2022; Kouzez, 2023; Olovede et al. 2021; Janot et al. 2016; Clemens et al. 2021; Thompson and Hay, 2015). It is known for its robustness (Chen and Xia, 2020) and is widely employed not only in social sciences (Margues, 2022; Kouzez, 2023; Oloyede et al. 2021) but also in natural sciences (Janot et al. 2016; Clemens et al. 2021; Thompson and Hay, 2015). Additionally, econometric software packages like EViews incorporate this method (EViews User Guide) for implementation, as it is pivotal for testing the endogeneity of equation regressors. Furthermore, it finds mention in widely used handbooks, such as the "Handbook of Financial Econometrics, Mathematics, Statistics, and Machine Learning" (Patrick, 2022a) and encyclopedias like the "Encyclopedia of Finance" (Patrick, 2022b).

The Durbin-Wu-Hausman specification test holds particular significance when dealing with panel data or constructing models with instrumental variables. The fundamental concept behind this test revolves around the evaluation of whether a specific independent variable within a regression model exhibits endogeneity, indicating a correlation with the error term. This test involves the computation of a test statistic, which is then contrasted with critical values from the chi-squared distribution to decide regarding the acceptance or rejection of the null hypothesis. We further employ the regressor endogeneity test. We employ the Durbin-Wu-Hausman specification test to examine the predictor variables in our model. The Durbin-Wu-Hausman specification test type tests can be employed to carry out least squares (Patrick, 2022a, 2022b). This test makes significant contributions to the advancement of specification tests that facilitate the comparison of parameter estimates obtained through diverse estimation methods. The test serves various objectives, including the selection between different model estimators (e.g., fixed effects versus random effects, Patrick, 2022b) and the assessment of alternative theoretical models (Chermak and Patrick, 2001, p. 78). The test is based on two estimators, providing:

$$x_t = X\beta + Z\Theta + u \tag{6}$$

$$X = G\Pi + \nu, Z = G\Delta + \eta \tag{7}$$

Where x_t is a vector of observations on a dependent variable, X is matrices of (possibly) endogenous explanatory variable for $x_t \in \mathbb{R}^{n \cdot m_x} \left(m_y + m_x = m > 1 \right)$, $G \in \mathbb{R}^{n \cdot l}$ is a matrix of exogenous instruments, and u_t is the vector of structural disturbances for $u_t = (u_1, \dots, u_n)$, $\in \mathbb{R}^n$, for the regressor z_t could be endogenous. If endogenous regressors are present, then the OLS estimators will supposedly fail this test. This is because it is assumed that the predictor factors and the error term are unrelated. All independent variables are assumed to be uncorrelated with the error term in OLS regression (Gujarati et al. 2012). The null hypothesis and alternative hypothesis of this test are follows:

$$H_0: cov(x_t, u_t) = 0 \tag{8}$$

$$H_1: cov(x_t, u_t) \neq 0 \tag{9}$$

The null hypothesis argues that the variable MOM_t , is exogenous. In the event that the null hypothesis is not rejected, it can be inferred that the ordinary least squares (OLS) estimators demonstrate consistency. Additionally, it may be inferred that the least squares estimator exhibits efficiency. The non-rejection of the null hypothesis implies that the primary variable may not exhibit significant endogeneity, and it could be more efficient to consider an alternative estimator like OLS.

The investor sentiment mechanism and causality analysis. Investor sentiment encompasses investors' collective outlook on the stock market, which can be shaped by a range of factors, including news, economic indicators, and market patterns. When investors grow more positive or negative in their market sentiment, it often leads to buying or selling of stocks. This, in turn, can result in short-term variations in stock prices and heightened market volatility. According to behavioral finance theory, investor sentiment significantly influences investment choices, asset valuations, and risk mitigation. Theoretical evidence supports the notion that investor sentiment can induce short-term fluctuations and even abrupt shifts in stock prices. Typically, sentiment affects companies characterized by youth, lack of profitability, high volatility, financial distress, growth orientation, small market capitalization, and non-payment of dividends. Positive (negative) shifts in sentiment result in decreases (increases) in return volatility and are linked to higher (lower) future excess returns. The impact of sentiments on stock returns is contingent on the prevailing market expectation. When the influence of sentiments on volatility is primarily unidirectional, it signifies that the pricepressure effect prevails in the market, benefiting noise traders during periods of elevated sentiment indices. Consequently, sentiment contributes to increased market volatility. Ultimately, heightened volatility in the stock market presents enhanced profit prospects for both long-term and short-term traders.

We conduct additional sentiment mechanism analysis for the following reasons. Firstly, heightened investor sentiment can significantly amplify stock market volatility. Furthermore, an upsurge in investor sentiment may induce selling pressure in the market, leading to reduced returns (Gao et al. 2022). Another factor to consider is that sentiment represents a systematic risk that is incorporated into pricing, resulting in contemporaneously positive correlations between shifts in sentiment and excess returns (Bouteska, 2020). Additionally, lower sentiment levels are associated with relatively higher subsequent returns, particularly in the case of smaller stocks, high volatility stocks, extreme growth stocks, and young stocks (Han, 2023). Lastly, the quest for information is positively linked to stock volatility and trading volume, as investors intensify their information searches to aid in decision-making and risk avoidance (Shu and Chang, 2015). Research on China's green stock markets has revealed that investor sentiment can significantly amplify stock volatility (Gao et al. 2022). Similarly, an examination of the impact of investor sentiment during the COVID-19 pandemic has shown that negative sentiment augments volatility, while positive sentiment diminishes it (Cevik et al. 2022). Furthermore, sentiment is considered a systematic risk that carries a price, with excess returns being simultaneously positively associated with shifts in sentiment (Lee et al. 2002). In emerging equity markets, it has been observed that an upsurge in investor sentiment can decrease returns, particularly influenced by the selling pressure prevailing in the market (Andleeb and Hassan, 2023). The objective of this approach is to capture, analyze, and understand the sentiment to inform stakeholders and improve decision-making processes within their domains (Song et al. 2022).

In the context of the Johannesburg stock exchange, research has indicated that when sentiment is low, subsequent returns tend to be relatively high for smaller stocks, high volatility stocks, extreme growth stocks, and young stocks (Muguto et al., 2022). Moreover, a study on investor sentiment during the COVID-19 pandemic has highlighted that investors intensify their information searches to aid their decision-making process in risk avoidance. This increased demand for information is positively linked to the volatility and trading volume of stocks (Jiang and Jin 2021). In summary, an elevation in investor sentiment can lead to a rise in stock market volatility. However, it is crucial to acknowledge that the relationship between investor sentiment and stock returns is intricate and can be influenced by a variety of factors, including the type of stock and prevailing market conditions.

To investigate the precise model through which the influence of star analysts impacts stock price fluctuations, we employ the investor sentiment mechanism. Our study assesses the effect of investor sentiment on stock market volatility by utilizing a GARCH (1, 1) model, wherein the investor sentiment proxy, represented by the VXN, is introduced as an exogenous regressor in the variance equation. The VXN index serves as a significant indicator of market expectations for short-term volatility, as evidenced by the pricing of options linked to the NASDAQ-100 Index. This metric quantifies the market's anticipation of shortterm volatility within the pricing of NASDAQ-100 options with a 30-day timeframe. To this end, we estimate a GARCH (1, 1) model as follows:

$$GARCH = C(2) + C(3) \cdot RESID(-1)^{2} + C(4) \cdot GARCH(-1) + C(5) \cdot NASDAQ VXN$$
(10)

Moreover, to investigate the extent of the impact of investor sentiment on stock market volatility, we adopt the frequencydomain causality test, as proposed by Breitung and Candelon (2006). The main motivation for utilizing this test is its capability to offer a more comprehensive view of the magnitude and direction of causation across various frequencies, encompassing both short-term and long-term (permanent) effects.

The traditional Granger causality test, utilizing a vector autoregressive (VAR) framework (Granger, 1969), is able to produce a single statistic that summarizes predictability across all frequencies. However, this conventional approach does not account for the possibility that causal relationships might vary across different frequencies, as noted by Geweke (1982) and Hosoya (1991). To address this limitation, Breitung and Candelon (2006) introduce a frequency domain Granger causality test as an alternative method to more accurately assess short, medium, and long-term Granger causality. The aim of this test is to provide a more precise understanding of causal links between variables. In our study, we employ a bivariate vector autoregressive (VAR) model to encompass a co-integrated system and a higher dimensional system. This approach allows for the separation of short-term and long-term predictability.

We employ a test to determine the existence of Granger causality at any given frequency (ϑ). This test examines the null hypothesis proposed by Geweke (1982), for $\Lambda_{y\to x}$ (ϑ) = 0, where Y does not Granger cause X (Özer et al. 2020), and which can be expressed as:

$$H_0: R(\mathfrak{d})\beta = 0, \tag{11}$$

with
$$\beta = (\beta_1, \dots, \beta_p)$$
 and $R(\vartheta) = \begin{pmatrix} \cos(\vartheta), \cos(2\vartheta), \dots, \cos(p\vartheta) \\ \sin(\vartheta), \sin(2\vartheta), \dots, \sin(p\vartheta) \end{pmatrix}$ (12)

When attempting to ascertain the results of the Breitung-Candelon (2006) Granger causality tests for the frequency (\mathfrak{d}) , it is necessary to compare the calculated test statistics value with the 5% chi-square significance value, considering 2 degrees of freedom.

Table 6 Performance of Long and Short Portfolios within each analyst group ((a) Panel 1. Long Portfolio, (b) Panel 2. Short Portfolio, and (c) Panel 3. Long-Short Portfolio).

(a) Panel 1. Long Portfolio						
	Star	Non-Star	Star-1	Star vs non-Star	Star-1 vs Star	Star-1 vs non-Star
	(1)	(2)	(3)	(4)	(5)	(6)
CAPM alpha FF 3-factor alpha Carhart 4-factor alpha	0.0146 ^{***} (6.56) 0.0145 ^{***} (6.51) 0.0145 ^{***} (6.47)	0.0151 ^{***} (24.52) 0.0151 ^{***} (24.49) 0.0151 ^{***} (24.50)	0.0177 ^{***} (7.65) 0.0175 ^{***} (7.57) 0.0174 ^{***} (7.52)	-0.0005 (-0.22) -0.0006 (-0.24) -0.0006 (-0.27)	0.0030 (0.95) 0.0030 (0.94) 0.0029 (0.92)	0.0025 (1.06) 0.0024 (1.03) 0.0023 (0.98)
(b) Panel 2. Short Portfolio	1					
	Star	Non-Star	Star-1	Star vs non-Star	Star-1 vs Star	Star-1 vs non-Star
	Star (1)	Non-Star (2)	Star-1 (3)	Star vs non-Star (4)	Star-1 vs Star (5)	Star-1 vs non-Star (6)

(c) Panel 3. Long-Short Portfolio						
	Star	Non-Star	Star-1	Star vs non- Star	Star-1 vs Star	Star-1 vs non- Star
	(1)	(2)	(3)	(4)	(5)	(6)
CAPM alpha	0.0232*** (8.15)	-0.00270 [*] (-2.06)	0.0270*** (6.57)	-	-	-
FF 3-factor alpha	0.0231 ^{***} (8.12)	-0.00274 [*] (-2.10)	0.0267*** (6.51)	-	-	-
Carhart 4-factor alpha	0.0229*** (8.07)	-0.00273* (-2.09)	0.0266*** (6.53)	-	-	-

t statistics in parentheses: *p < 0.05, **p < 0.01, ***p < 0.001.

Note: Alphas are calculated for each group of analysts, i.e., stars and non-stars, as an intercept from the regression. As an additional test, the table includes alphas for the Star-1 category, those Star analysts who are ranked "number one". The right side of the table shows the difference in alphas between each group of analysts. The long-short portfolio suggests that stars, on average, have higher alphas compared to non-stars. "Number one"-ranked stars show overperformance in all portfolios.

Results and discussion

Factors' model results. Table 6A-C display risk-adjusted returns (alphas) calculated using CAPM (1965), Fama-French 3 factor model (1993), and Carhart 4 factor model (1997) for long and short portfolios within each analyst group. In addition to the overall sample of analysts categorized as Star and Non-Star, Table 6A-C encompasses a subset known as Star-1 analysts, who have achieved the top rating according to StarMine's rankings. The table has been divided into three distinct sections, denoted as Panel 1 (6 A), Panel 2 (6B), and Panel 3 (6C). Panel 1 represents the long portfolio, Panel 2 represents the short portfolio, and Panel 3 displays the disparity between the long and short portfolios. The table categorizes analysts into three distinct groups, namely Stars, Non-Stars, and Star-1, and provides information about the alpha differentials associated with each group. In Table 6A-C, it is evident that the regression findings demonstrate highly significant alpha across all panels and analyst groups. The presence of substantial alphas implies that analysts' recommendations, in spite of categorization into Stars and Non-Stars, can yield abnormal returns, hence rejecting the efficient market hypothesis (EMH). These findings are in line with works of Womack (1996), Barber et al. (2001), and Fang and Yasuda (2014).

Table 6A, represents the average daily alphas for the long portfolio. In general, the results of the regressions show strongly significant excess returns for each group of analysts. Specifically, CAPM shows that Star analysts generate 0.0146% average daily (0.3066% monthly) excess returns. On first glance, non-stars outperform stars, with average 0.0151% daily (0.3171% monthly) abnormal returns. Coefficients are not significantly changed when different models are applied. As such, the Fama-French 3-factor model and the Carhart 4-factor model indicate approximately the same level of alphas as in CAPM. Looking at the right side of Table 6A, the results of the parametric test show insignificant alpha differentials across all the groups of analysts. Particularly,

Kucheev and Sorensson (2019), report significant evidence that Stars can generate on average 0.35% monthly excess returns for Long Portfolio. These findings are approximately similar with the results obtained in our study. Despite the higher levels of alpha among non-stars and the inconsistency with most of the previous literature (e.g., Fang and Yasuda, 2014), it is hard to conclude that rankings are "popularity contests" (Emery and Li, 2009), since the result of parametric test shows insignificant difference between analyst groups. The reason for insignificant outperformance of Non-Stars in Long Portfolio might be due to increased attention on high weight firms that constitute the NASDAQ 100 index, as described in Byun and Roland (2020). Another explanation to the differences of reputation effect between the Short and Long Portfolios might be the "positivity bias" (Barber et al. 2001; Lötter and vd M Smit, 2018).

The short portfolio (Table 6B) is notably different from the long portfolio. Specifically, the results of three regression models suggest that stars have significantly higher excess returns compared to non-stars. As such, by shorting stocks in the short portfolio, star analysts generate average -0.0086% daily (0.1806% monthly) abnormal returns, while non-star analysts generate 0.0178% daily (0.3738% monthly) abnormal returns. It is vital to note that a short portfolio is considered profitable when the abnormal returns come with a negative sign, since shorting is done in the opposite direction. As a result, parametric test results indicate that star analysts significantly outperform non-stars by an average of -0.0265% daily (-0.5565% monthly). Coefficients and t-stats are unchanged when different regression models are applied. The statistically significant (at 1% level) monthly 0.5565% alpha differential is fully consistent with the results obtained by Fang and Yasuda (2014).

Table 6C represents the differences between long and short portfolios. Consistent with previous portfolio results, the excess returns in the long-short portfolio are approximately equal across all three models. Hence, focusing on the Carhart 4-factor alphas, stars generate on average 0.0229% daily (0.4809% monthly) abnormal returns. Calculations for alphas in the Stars group are statistically significant at the 1% level. The Non-Stars' long-short portfolio shows -0.00270% daily (-0.0567% monthly) abnormal returns, which are significant at the 10% level. While the abnormal returns generated by the long-short portfolio of stars match the results obtained by Kucheev and Sorensson (2019), non-stars are slightly different. More precisely, it finds that stars generate 0.53% monthly abnormal returns, while non-stars show 0.47% monthly alphas.

The study conducts a supplemental test to identify whether stars tend to underperform non-stars in long portfolios under any conditions. For these purposes, the study separates the "number one" ranked stars (Star-1) from the Stars group to construct the same models with alphas generated only by the current group of stars. Table 6A-C, Column 3, represents the alphas obtained from different asset pricing models for the Star-1 group. Along with alphas, columns 5-6 display alpha differentials between Star-1 vs. stars and Star-1 vs. non-stars, respectively. From the Carhart 4-factor alphas of the long portfolio, it can be seen that "number one" ranked stars generate on average 0.0174% daily (0.3717% monthly) excess returns, which are statistically significant at the 1% level. Despite the high levels of alpha generated in Long Portfolio, parametric test results for the differences between Stars-1 and other groups remain statistically insignificant. The short portfolio constructed on the recommendations of the Star-1 group shows the highest alphas (statistically significant at the 5% level) compared to other analyst groups. As such, the Star-1 group generates -0.0092% daily (0.1932% monthly) excess returns.

The difference between Star-1 and Star Short Portfolio is 0.0147% monthly, and it is statistically insignificant. However, the difference between the Star-1 and Non-Star Short Portfolios indicates highly statistically significant of 0.567% monthly abnormal returns. Analogously, in the long-short portfolio, the Star-1 group shows the highest and most statistically significant alphas in all three asset pricing models (e.g., 0.0266% daily and 0.5586% monthly alphas from the Carhart 4-factor model). The results obtained for "number one" ranked stars are in the line with similarly mentioned studies. For instance, Fang and Yasuda (2014) divided the II sample into top-rank AAs (like Star-1) and found that "number one" ranked stars generate statistically significant and high (2.91% monthly in the Long Portfolio) abnormal returns. However, the difference between Star-1 and non-Star groups is consistent only in the short portfolio, since the long portfolio differentials are not statistically significant in our study. The results of the long-short portfolio in the Star-1 subgroup of analysts are also somehow consistent with the results obtained by Kucheev and Sorensson (2019).

The seemingly equal performance of stars and non-stars in the long portfolio can be explained by the positive nature of "strong buy" and "buy" recommendations. It means that market participants prefer to obtain positive signals and act accordingly. On the other hand, it can be challenging for the analyst to provide negative recommendations (Emery and Li, 2009). Extended access allows the analysts to be aware of significant news and announcements much earlier than non-star analysts. It also allows them to process the non-publicly available information, while non-stars rely on the single piece of information that is publicly available. Moreover, the calculations of the non-star analysts might include only publicly available fundamental information, while the stars might gain superior fundamental knowledge. One possible explanation for the prevalence of more lucrative "sell" and "strong sell" recommendations among star analysts is their superior reputation and expertize in the field. Kadan et al.

(2020) emphasize the significance of benchmarks as foundational elements for the recommendations. The authors underscore the criticality of understanding the source and characteristics of the information that support the analyst's findings.

Relative to the investigation by Kucheev and Sorensson (2019), our analysis does incorporate temporal considerations, such as monthly alpha generation. However, Kucheev and Sorensson (2019) delve into the seasonality of analyst optimism, uncovering a lack of significant relationship between seasonal target prices and market returns. They identify an optimism cycle that correlates with the rhythm of earnings disclosures rather than seasonal variations. Additionally, our findings suggest that analysts with a star ranking outperform the market, albeit predominantly in the short term. Kucheev and Sorensson (2019) propose that investors could gain from recognizing these cycles of optimism, hinting at potential market inefficiencies if such biases from analysts are predictable and correspond with the timing of corporate disclosures.

In concert with the conclusions drawn by Su et al. (2019), our study did not uncover a consistent benefit from analyst upgrades or downgrades when considering the market as a whole. This indicates that an analyst's reputation or brokerage rank is not invariably indicative of the value of their recommendations, particularly when transaction costs are considered. Our research delineates a clear demarcation between star analysts and their non-star counterparts in their capacity to secure abnormal returns, with this phenomenon being particularly pronounced in short-term portfolios. In alignment with Su et al. (2019), we also identify specific industry contexts—in this case, high-tech sectors—where analyst insights appear to offer genuine profit potential, thereby enhancing our nuanced comprehension of the contexts in which analyst recommendations may be of value.

Our findings corroborate the link between analyst rankings and investment performance, in accordance with Fang and Yasuda (2014), but the distinction is not significant within long-term portfolios—merely in short-term ones—implying a more subtle influence. On the matter of market efficiency, our results echo Fang and Yasuda's (2014) evidence that certain analysts deliver consistently valuable insights that lead to abnormal returns. We suggest, however, that such effects are more emphatically observed in short-term portfolios. Meanwhile, Fang and Yasuda (2014) explore the overarching capabilities of analysts with high reputations to excel over various investment timeframes.

Endogeneity test results. In assessing whether an endogeneity concern between the momentum factor and other included variables in our statistical model existed, we implemented the Durbin-Wu-Hausman assessment. The statistical outputs from this scrutiny are concisely summarized in the provided Table 7. The essence of this test is to compare the ordinary least squares (OLS) estimates with those procured through instrumental variable (IV) techniques, upholding the null hypothesis which posits that the suspected endogenous variables are actually exogenous.

The data reported in Table 7 reveal that the p-value linked to the J-statistic is quantified at 0.7537, which markedly exceeds the commonly accepted 5% significance benchmark. This significant statistic supports the maintenance of the null hypothesis, thereby substantiating the external position of the momentum variable within our specified econometric arrangement. The prominently high p-value suggests that the estimations employed are not contaminated by biases typically introduced by endogeneity, hence they are deemed both efficient in nature and consistent in their estimative capability—a vital prerequisite for authentic econometric analysis.

Table 7 The outcomes	of the endoger	nous testir	ıg.
	Value	df	Probability
Difference in J-stats	0.0985 Value	1	0.7536
Restricted J-statistic Unrestricted J-statistic	0.0985 0		

The entry labeled 'Difference in J-stats' records a value of 0.0985 with a p-value that mirrors this at 0.7536, cementing the exogeneity stance for the momentum factor. Further substantiating this conclusion is the fact that the 'Restricted J-statistic' and 'Unrestricted J-statistic' mirror each other in value, fortifying the strength of the claim for exogeneity. This investigation not only bolsters the methodological foundation of our current analysis but also provides a template for the architectural design of future research models. By establishing the exogeneity of the momentum factor, the research empowers further scholarly pursuits to build upon econometric frameworks that assure structural integrity and methodological reliability.

The investor sentiment and causality results. Our study ventures to gauge the impact of investor mood on equity returns by using the Volatility Index (VXN) as a surrogate indicator of sentiment in the marketplace. The analytical findings, as outlined in Table 8, indicate that both the Autoregressive Conditional Heteroskedasticity (ARCH) and the Generalized ARCH (GARCH) coefficients are not only statistically notable and carry a positive sign, but they also collectively sum up to less than unity. This implies a clear and constructive relationship between investor sentiment levels and the oscillations in stock returns; a surge in sentiment leads to heightened volatility in the stock market, as depicted in Fig. 1.

In delving deeper into how frequently investor sentiment influences market volatility, our research incorporates the frequency-domain causality framework articulated by Breitung and Candelon (2006). This methodological approach sheds light on the causality's direction and magnitude across different time horizons. The insights from this methodology are visualized in Figs. 2 and 3, where the frequency parameter (a) on the horizontal axis serves to determine the cycle's length in days through the equation:

$$T = 2\pi/\mathfrak{o} \tag{12}$$

The vertical axis in Figs. 2 and 3 illustrates the p-values of the calculated test statistics (F-statistics). Each test statistic and its associated p-values are obtained across the frequency range 0 < $\varepsilon < \pi$. Also, by using the (ε), the study defines the long-run as $\varepsilon = 0.01$ and short-run as $\varepsilon = 2.5$.

According to Fig. 2, investor sentiments can contribute not only to the short run but also in the long run to stock (ω) (Özer et al. 2020). Interpreting the graphical analysis in Fig. 2 reveals that investor sentiment exerts influence not solely in immediate time frames but extends its reach into more extended periods as well. Conversely, Fig. 3 intimates that fluctuations in the stock market provoke shifts in investor sentiment predominantly within a window of 5 to 10 days, albeit these fluctuations do not have permanent effects.

The most profound contribution of our investigation lies in the finding that investor sentiment plays a pivotal role in the volatility of stock returns. This insight builds upon and extends the corpus of existing literature, providing fresh perspectives on the significance of market sentiment.

Table 8 The	Table 8 The GARCH (1,1) estimates.								
Variable	Coefficient	Std. Error	z-Statistic	Prob.					
С	-0.0049	0.0008	-6.017	1.770e-09					
	Variance Equation								
С	-0.001388	0.000146	-9.490048	0.0000					
RESID(-1) ²	0.381635	0.030422	12.54492	0.0000					
GARCH(-1)	0.082328	0.024061	3.421677	0.0006					
NASDAQ	0.000191	1.06e-05	18.12596	0.0000					

Mechanism. The intricate relationship among the prognostications of renowned analysts, the inherent instability of the market, and the collective mood of investors intricately intertwines to mirror the multifaceted nature of financial markets and investor conduct. As demonstrated in Fig. 4, the counsel dispensed by preeminent analysts wields the potential to sway overall market sentiment and precipitate fluctuations in the short-term market climate. For example, an endorsement categorized as a "buy" or "strong buy" can foster a buoyant outlook on a firm's future, thereby elevating its stock value. Conversely, a directive to "sell" or "strong sell" can have an antithetical effect, leading to a dip in stock prices.

There are exceptional instances where the counsel from these esteemed analysts, coupled with prevailing market sentiment, can significantly intensify price dynamics and volatility within the marketplace—a phenomenon colloquially termed as "herding behavior." This observation confirms a cyclical interplay where top-tier analyst advice informs market sentiment, which in turn feeds back into market volatility. It is recognized that adverse sentiment amongst investors is a precursor to greater market turbulence when juxtaposed with optimistic sentiment. Therefore, the guidance from leading analysts, the fickleness of the market, and the sentiment among investors are dynamically linked. This triadic relationship forges cyclical interactions that can exacerbate the ebb and flow of market movements. Thus, a comprehensive grasp of these interrelations is imperative for investors who strive to base their financial decisions on deep market insight.

Figure 4 underscores this interconnectivity, highlighting the significant nexus between the insights of star analysts, market volatility, and investor sentiment.

Prior literature in the field has predominantly concentrated on isolated aspects of analyst influence, either focusing on the direct impact of recommendations on stock performance or on the behavioral response of the market to such recommendations. However, our study transcends the conventional analysis by illustrating the cyclical nature of the influence that star analysts exert on market sentiment and, concomitantly, how these sentiments synergistically amplify market volatility. The innovative proposition of our research is the identification of a dual feedback loop, which represents a significant departure from existing paradigms. It postulates that the recommendations from star analysts instigate shifts in market sentiment, which then precipitate short-term market volatility. In turn, this volatility influences subsequent market sentiment, establishing a dynamic and continuous loop of influence. This conceptual framework captures the nuanced reality that negative investor sentiments, often undervalued in existing literature, precipitate heightened market volatility when compared to positive sentiments.

Furthermore, by recognizing the herding behavior that occasionally amplifies the impact of analysts' recommendations, this study contributes to a refined understanding of market overreactions and corrections. The notion that these interactions can lead to feedback loops offers a new perspective on how market actors might exploit systematic patterns in market responses to strategic advantage.

Conclusion

The present study provides empirical evidence supporting the utility of analyst recommendations in the process of stock selection. The



Fig. 1 Stock market volatility. This figure represents stock market volatility. The vertical axis represents conditional standard deviation for each observation. The horizontal axis represents the observations.



Fig. 2 Causality test analysis: Investor sentiment No Cause-Granger stock return volatility. This figure illustrates results of causality test analysis for investor sentiment no Cause-Granger stock return volatility. The horizontal axis represents the frequency parameter (*ω*). The vertical axis illustrates the p-values of the calculated F test statistics.

findings of the study offer empirical support for the notion that analysts with star rankings have the ability to generate superior alphas, hence surpassing the average performance of the market. The study also highlights the effectiveness of the ranking system used by StarMine in identifying the star analysts with superior stock-picking abilities. The signaling theory assumes that all the information on company's financial health is not available to all the investors at the same time Thus, the study findings indicate that signaling theory has existed in the financial market for a short period of time. However, it diminishes as analyst recommendations are incorporated into market prices, closing the information gap.

The present study's findings have many practical implications, notably highlighting the potential benefits for market participants who integrate analyst suggestions into their investing decisionmaking procedures. It also shows that market participants should conduct due diligence on analyst recommendations to generate higher returns. Then market participants can consider Star Analyst as a benchmark to compare their investment performance. The study findings indicate that there is no significant distinction in alphas between the star and non-star in the long portfolio. It supports EMH that all available information is quickly and accurately reflected in asset prices. The study results also indicate that information symmetry does play a role. However, by implementing the same investment strategy, star analysts cannot beat the market continuously due to increased market competition. Even though star analysts possess superior skills, the widespread accessibility of advanced technical tools for analysis diminishes the overall disparities in skill levels among analysts when it comes to making wellinformed financial decisions. It is difficult to continuously beat the market with the same strategy according to EMH. The study results also indicate that information symmetry does play a role, and star analyst recommendations play a vital role in mitigating information asymmetry in financial markets.

The study's limitations are primarily associated with its data sources. The analysis of analyst rankings in this research was based on a specific set of data. As indicated in Section 2.3, broadening the variety of data sources could offer deeper and more generalizable insights. Including newer data and more companies in dataset, could thus enhance the study's current relevance and applicability (like NYSE or S&P market data).



Fig. 3 Causality test analysis: stock return volatility No Cause-Granger Investor sentiment. This figure illustrates results of causality test analysis for stock return volatility no Cause-Granger Investor sentiment. The horizontal axis represents the frequency parameter (*b*). The vertical axis illustrates the p-values of the calculated F test statistics.



Fig. 4 The interconnectedness between star analyst recommendations, volatility, and market sentiment.

This research addresses the impact of information asymmetry in financial markets. Subsequent studies could delve deeper into this topic, specifically investigating the role of top-tier analysts in narrowing information disparities and how this affects overall market efficiency. Additionally, leveraging the foundational use of the GARCH model and frequency-domain causality analysis in this study, future research might probe further into the dynamics between investor sentiment and stock performance. This exploration could extend to analyzing the effects of various news types and economic indicators on investor actions and financial market results. Policymakers may consider different measures to enhance transparency, accuracy, and accountability in analyst recommendations to protect investors' interests.

Data availability

The datasets are available from the corresponding author on reasonable request.

Received: 15 May 2023; Accepted: 11 December 2023; Published online: 02 January 2024

Notes

- 1 https://mba.tuck.dartmouth.edu/pages/faculty/ken.french/data_library.html
- 2 Portfolio creation and evaluation include a variety of approaches and considerations. For instance, Vukovic et al. (2022a) note that some research suggests using an inflation allocation line. The neural network method is used to calculate and forecast portfolio returns in the study by Li et al. (2017b). The research of Vukovic et al. (2022b) then employs support vector machines, group techniques of data management, long shortterm memory, and Markov switching autoregression for future value projections.

References

- Andleeb R, Hassan A (2023) Impact of investor sentiment on contemporaneous and future equity returns in emerging markets. SAGE Open 13(3):21582440231193568
- Balakrishnan A, Maiti M (2017) Dynamics of size and value factors in stock returns: Evidence from India. Indian J Financ 11(6):21–35. https://doi.org/10. 17010/ijf/2017/v11i6/115593
- Barber BM, Lehavy R, Trueman B (2007) Comparing the stock recommendation performance of investment banks and independent research firms. J Financial Econ 85(2):490–517. https://doi.org/10.1016/j.jfineco.2005.09.004
- Barber BM, Lehavy R, Trueman B (2010) Ratings changes, ratings levels, and the predictive value of analysts' recommendations. Financial Manag 39(2):533-553. https://doi.org/10.1111/j.1755-053x.2010.01083.x
- Barber BM, Lehavy R, McNichols M, Trueman B (2001) Can investors profit from the prophets? Security analyst recommendations and stock returns. J Financ 56(2):531–563. https://doi.org/10.1111/0022-1082.00336
- Barber BR, Lehavy R, McNichols M, Trueman B (2006) Buys. Holds, and Sells: The distribution of investment banks' stock ratings and the implications for the profitability of analysts' recommendations. J Account Econ 41(1-2):87–117. https://doi.org/10.1016/j.jacceco.2005.10.001

- Bouteska A (2020) Some evidence from a principal component approach to measure a new investor sentiment index in the Tunisian stock market. Managerial Finance 46(3):401–420
- Breitung J, Candelon B (2006) Testing for short-and long-run causality: A frequency-domain approach. J Econ 132(2):363–378. https://doi.org/10.1016/ j.jeconom.2005.02.004
- Brown KC, Harlow WV, Tinic S (1993) The risk and required return of common stock following major price innovations. J Financial Quant Anal 28(1):101–116. https://doi.org/10.2307/2331153
- Byun S, Roland KC (2020) Analyst reputation and limited attention: How does firm visibility impact measures of reputation? Asia-Pac J Account Econ 29(3):603–621. https://doi.org/10.1080/16081625.2020.1770614
- Carhart M (1997) On persistence in mutual fund performance. J Financ 52(1):57-82
- Cevik E, Kirci Altinkeski B, Cevik EI, Dibooglu S (2022) Investor sentiments and stock markets during the COVID-19 pandemic. Financial Innov 8(1):69
- Chen Z, Xia H (2020) Trend instrumental variable regression with an application to the US New Keynesian Phillips Curve. Economic Modeling 93:595-604
- Chermak JM, Patrick RH (2001) A microeconometric test of the theory of exhaustible resources. J Environ Econ Manag 42(1):82-103
- Clemens A, Mishra PP, Eirini F, Terho L, Raitakari OT, Liisa K-J, Katriina H (2021) C-reactive protein and temperament. Instrum variable Anal Brain, Behav, Immun - Health 14:100241–100241
- De Bondt WF, Thaler RH (1985) Does the stock market overreact. J Financ 40(3):793–808. https://doi.org/10.1111/j.1540-6261.1985.tb05004.x
- Dutta A (2015) Improved calendar time approach for measuring long-run anomalies. Cogent Econ Financ 3(1):1065948. https://doi.org/10.1080/ 23322039.2015.1065948
- Ertimur Y, Mayew WJ, Stubben SR (2011) Analyst reputation and the issuance of disaggregated earnings forecasts to I/B/E/S. Rev Account Stud 16:29–58. https://doi.org/10.1007/s11142-009-9116-5
- Emery D, Li X (2009) Are the wall street analyst rankings popularity contests? J Financial Quant Anal 44(2):411–437. https://doi.org/10.1017/ s0022109009090140
- Fama EF (1965) The behavior of stock-market prices. J Bus 38(1):34-105. https:// doi.org/10.1086/294743
- Fama EF, French K (1993) Common risk factors in the returns on stocks and bonds. J Financial Econ 33(1):3–56. https://doi.org/10.1016/0304-405x(93)90023-5
- Fang L, Yasuda A (2014) Are stars' opinions worth more? The relation between analyst reputation and recommendation values. J Financial Serv Res 46(3):235–269. https://doi.org/10.1007/s10693-013-0178-y
- Fang LH, Yasuda A (2009) The effectiveness of reputation as a disciplinary mechanism in sell-side research. Rev Financial Stud 22(9):3735–3777. https:// doi.org/10.1093/rfs/hhn116
- Gao H, Liu Y, Zhong L (2022) Investor sentiment and stock market volatility: Evidence from China's green stock markets. J Asia-Pac Bus 17(2):173-202
- Geweke J (1982) Measurement of linear dependence and feedback between multiple time series. J Am Stat Assoc 77(378):304–313. https://doi.org/10.2307/2287238
- Granger CWJ (1969) Investigating causal relations by econometric models and crossspectral methods. Econometrica 37(3):424–438. https://doi.org/10.2307/ 1912791
- Gujarati DN, Porter DC, Gunasekar S (2012) Basic Econometrics. McGraw Hill Education (India) Private Limited, New York, NY
- Guo L, Li FW, Wei KJ (2020) Security analysts and capital market anomalies. J Financial Econ 137(1):204–230. https://doi.org/10.1016/j.jfineco.2020.01.002
- Hou K, Xue C, Zhang L (2015) Digesting anomalies: an investment approach. Rev Financial Stud 28(3):650-705
- Han Y (2023) The Influence of Investor Sentiment on Stock Volatility: Empirical Tests from the Financial Social Media Platform
- Hobbs J, Singh V, Chakraborty M (2021) Institutional underperformance: Should managers listen to the sell-side before trading? Rev Quant Financ Account 57(1):389–410. https://doi.org/10.1007/s11156-020-00948-z
- Hosoya Y (1991) The decomposition and measurement of the interdependency between second-order stationary processes. Probab Theory Relat Fields 88(4):429–444. https://doi.org/10.1007/BF01192551
- Janot A, Vandanjon PO, Gautier M (2016) A revised Durbin-Wu-Hausman test for industrial robot identification. Control Eng Pract 48:52–62
- Jagadeesh V, Gehrke J, Labrinidis A, Papakonstantinou Y, Patel JM, Ramakrishnan R, Shahabi C (2014) Big Data and Its Technical Challenges. Communications of the ACM 57(7):86–94. https://doi.org/10.1145/2611567
- Jegadeesh N, Titman S (1993) Returns to Buying Winners and Selling Losers: Implications for Stock Market Efficiency. *The. J Financ* 48(1):65–91
- Jegadeesh N, Kim W (2006) Value of analyst recommendations: International evidence. J Financial Mark 9(3):274–309. https://doi.org/10.1016/j.finmar. 2006.05.001
- Jiang S, Jin X (2021) Effects of investor sentiment on stock return volatility: A spatio-temporal dynamic panel model. Econ Model 97:298–306

Kadan O, Madureira L, Wang R, Zach T (2020) Sell-side analysts' benchmarks. Account Rev 95(1):211-232. https://doi.org/10.2308/accr-52446

Kessler A(2001) We are all analysts now The Wall Street J 30:A18, https://www.wsj. com/articles/SB996440841790888620

- Kim MS, Zapatero F (2023). Uncertainty and Dispersion of Opinions. Available at SSRN, 1569888. https://doi.org/10.2139/ssrn.1569888
- Kothari SP, So E, Verdi R (2016) Analysts' forecasts and asset pricing: a survey. Annu Rev Financial Econ 8(1):197–219. https://doi.org/10.1146/annurevfinancial-121415-032930
- Kouzez M (2023) Political environment and bank performance: Does bank size matter? Econ Syst 47(1):101056–101056
- Kucheev Y, Ruiz F, Sorensson T (2017) Do stars shine? Comparing the performance persistence of star sell-side analysts listed by institutional investor, the wall street journal, and StarMine. J Financ Serv 52:277–305
- Kucheev Y, Sorensson T (2019) The seasonality in sell-side analysts' recommendations. Financ Res Lett 29:162–168. https://doi.org/10.1016/j.frl.2018. 07.001
- Lee WY, Jiang CX, Indro DC (2002) Stock market volatility, excess returns, and the role of investor sentiment. J Bank Financ 26(12):2277-2299
- Leone AJ, Wu JS (2007) What does it take to become a superstar? Evidence from institutional investor rankings of financial analysts. Evidence from Institutional Investor Rankings of Financial Analysts (May 23, 2007). Simon School of Business Working Paper No. FR, 02-12. https://doi.org/10.2139/ssrn. 313594
- Li D, Chen Z, An Z, Murong M (2017b) Do financial analysts play a role in shaping the rival response of target firms? International evidence. J Corp Financ 45:84–103. https://doi.org/10.1016/j.jcorpfin.2017.04.011
- Li RYM, Fong S, Chong KWS (2017a) Forecasting the REITs and stock indices: group method of data handling neural network approach. Pac Rim Prop Res J 23(2):123–160
- Liu J, Stambaugh RF, Yuan Y (2019) Size and value in China. J financial Econ 134(1):48–69. https://doi.org/10.1016/j.jfineco.2019.03.008
- Loh R, Stulz R (2011) When are analyst recommendation changes influential? Rev Financial Stud 24(2):593–627. https://doi.org/10.1093/rfs/hhq094
- Lötter R, vd M Smit E (2018) Recommendation pattern preferences and share analysts bias on the JSE. Stud Econ Econ 42(1):1–22. https://doi.org/10.1080/ 10800379.2018.12097324
- Maiti M (2019) OLS versus quantile regression in extreme distributions. Contaduría y administración 64(2):1–11. https://doi.org/10.22201/fca.24488410e. 2018.1702
- Maiti M, Balakrishnan A (2018) Is human capital the sixth factor? J Economic Stud 45(4):710–737. https://doi.org/10.1108/JES-05-2017-0132
- Maiti M, Balakrishnan A (2020) Can leverage effect coexist with value effect? IIMB Manag Rev 32(1):7–23. https://doi.org/10.1016/j.iimb.2019.07.011
- Marques AM (2022) Is income inequality good or bad for growth? Further empirical evidence using data for all Brazilian cities. Struct Change Economic Dyn 62:360–376
- Mulligan RF (1996) Export-import endogeneity in the context of the Thirlwall-Hussain model: an application of the Durbin-Wu-Hausman test incorporating a Monte Carlo experiment. Appl Econ Lett 3(4):275–279
- Muguto HT, Muguto L, Bhayat A, Ncalane H, Jack KJ, Abdullah S, Muzindutsi PF (2022) The impact of investor sentiment on sectoral returns and volatility: Evidence from the Johannesburg stock exchange. Cogent Econ Finance 10(1):2158007
- Oloyede BM, Evans O, Ejemeyovwi JO (2021) Trade openness and economic growth in Africa's regional economic communities: empirical evidence from ECOWAS and SADC. Heliyon 7(5):e06996–e06996
- Özer M, Kamenković S, Grubišić Z (2020) Frequency domain causality analysis of intra-and inter-regional return and volatility spillovers of South-East European (SEE) stock markets. Economic Res-Ekonomska istraživanja 33(1):1–25. https://doi.org/10.1080/1331677X.2019.1699138
- Patrick RH (2022a) Financial Panel Data Models, Strict Versus Contemporaneous Exogeneity, and Durbin-Wu-Hausman Specification Tests. In: Encyclopedia of Finance. Springer, p 1799–1828
- Patrick RH (2022b) Durbin–Wu–Hausman Specification Tests. In Handbook of Financial Econometrics, Mathematics, Statistics, and Machine Learning. World Scientific, p 1075–1108
- Refinitiv (2023) Is Thomson Reuters terminal database, collected in February 2023, collected from this terminal
- Sharpe WF (1964) Capital asset prices: a theory of market equilibrium under conditions of risk. J Financ 19(3):425–442. https://doi.org/10.2307/2977928
- Shu HC, Chang JH (2015) Investor sentiment and financial market volatility. J Behav Financ 16(3):206–219
- Song L, Li RYM, Yao Q (2022) An informal institution comparative study of occupational safety knowledge sharing via French and English Tweets: Languaculture, weak-strong ties and AI sentiment perspectives. Saf Sci 147:105602

- StarMine (2020) Is database collected from Refinitiv Thomson Reuters terminal database
- Stickel S (1995) Anatomy of the performance of buy and sell recommendations. Financial Analysts J 51(5):25–39. https://doi.org/10.2469/faj.v51. n5.1933
- Su C, Zhang H, Bangassa K, Joseph NL (2019) On the investment value of sell-side analyst recommendation revisions in the UK. Rev Quant Financ Account 53(1):257–293. https://doi.org/10.1007/s11156-018-0749-y
- Sun S, Li T, Ma H, Li RYM, Gouliamos K, Zheng J, Yue XG (2020) Does employee quality affect corporate social responsibility? Evidence from China. Sustainability 12(7):2692
- Thompson C, Hay JW (2015) Estimating the association between metabolic risk factors and marijuana use in U.S. adults using data from the continuous National Health and Nutrition Examination Survey. Ann Epidemiol 25(7):486-491
- Vukovic D, Maiti M, Frömmel M (2022a) Inflation and portfolio selection. Financ Res Lett 50(Dec 2022):103202. https://doi.org/10.1016/j.frl.2022. 103202
- Vukovic DB, Ugolnikov V, Maiti M (2020) Analyst Says a Lot but Should You Listen: Evidence from Russia. J Economic Stud 47(4):729–745. https://doi. org/10.1108/jes-10-2018-0352
- Vukovic DB, Ugolnikov V, Maiti M (2021) Sell-side analysts' recommendations a value or noise. Int J Financ Econ 26(2):3134–3151 10.1Yasar002/ ijfe.1954
- Vukovic DB, Ingenito S, Maiti M (2023) Time series momentum: Evidence from the European equity market. Heliyon 9:e12989. https://doi.org/10.1016/j. heliyon.2023.e12989
- Vukovic DB, Romanyuk K, Ivashchenko S, Grigorieva E (2022b) Are CDS spreads predictable during the Covid-19 pandemic? Forecasting based on SVM, GMDH, LSTM and Markov switching autoregression. Expert Syst Appl 194:116553. https://doi.org/10.1016/j.eswa.2022.116553
- Womack K (1996) Do brokerage analysts' recommendations have investment value? J Financ 51(1):137–167. https://doi.org/10.1111/j.1540-6261.1996. tb05205.x
- Yasar B, Martin T, Kiessling T (2020) An empirical test of signaling theory. Manag Res Rev 43(1):1309–1335. https://doi.org/10.1108/MRR-08-2019-0338

Acknowledgements

The work of author DBV for his contribution in the paper (investigation of market efficiency through investor sentiment, mechanism, and volatilities) has been supported by the RSF grant 24-28-00521 for the project "Market efficiency in turmoil: an arbitrage opportunity and arelative optimization in the long run" (https://rscf.ru/en/project/24-28-00521/).

Author contributions

Conceptualization, O.K., D.V., M.M; data curation, O.K., M. Ö and D.V.; formal analysis, D.V., O.K., M.M., and M.Ö.; investigation, O.K., M.Ö., D.V.; methodology, O.K., D.V., and M.Ö.; resources, O.K., M.R.; software, O.K, and M.Ö.; supervision, D.V.; validation, D.V., O.K., M.R., and M.Ö.; visualization, O.K., M.Ö. and M.R.; writing-original draft, O.K., D.V.; writing-review and editing, O.K., D.V., M.M., and M.Ö., project administration, D.V.

Competing interests

The authors declare no competing interests.

Ethical approval

This article does not contain any studies with human participants performed by any of the authors.

Informed consent

This article does not contain any studies with human participants performed by any of the authors.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1057/s41599-023-02527-8.

Correspondence and requests for materials should be addressed to Darko B. Vukovic.

Reprints and permission information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2024